\documentclass{adb}
\usepackage{graphicx}


\begin{document}

% ****************** TITLE ****************************************

\title{Assignment 2 - <DBMS NAME>}

% ****************** AUTHORS **************************************

\numberofauthors{3} 

\author{
% 1st. author
\alignauthor Student One
% 2nd. author
\alignauthor Student Two
% 3rd. author
\alignauthor Student Three
}


\maketitle

\section{Concurrency control mechanisms}

Describe briefly the concurrency control mechanisms supported by your DBMS.

\section{Locking modes}

What lock modes are supported by your DBMS (shared, exclusive, others)?

\section{Resource locking}

Which resource types can be locked in your DBMS (schema, tables, rows, others)?

Show on simple examples how (if possible) to explicitly lock resources (for example, tables and tuples)?

\section{Booking experiment}

Put here experiment results and conclusions.

\section{Viewing locks [optional]}

\end{document}
